Author's response to reviews

Title: Health care for immigrants in Europe: is there still consensus among country experts about principles of good practice? A Delphi study

Authors:

Walter Devillé (w.deville@nivel.nl)
Tim Greacen (tgreacen@ch-maison-blanche.fr)
Marija Bogic (m.bogic@qmull.ac.uk)
Marie Dauvrin (marie.dauvrin@uclouvain.be)
Sónia Dias (SFDias@ihmt.unl.pt)
Andrea Gaddini (gaddini@asplazio.it)
Natasja Koitzsch Jensen (naje@sund.ku.dk)
Christine Karamanidou (ckaramanidou@gmail.com)
Ulrike Kluge (Ulrike.Kluge@charite.de)
Ritva Mertaniemi (ritva.mertaniemi@thl.fi)
Rosa Puigpinós i Riera (ruigpi@aspb.cat)
Attila Sárváry (asarvary@de-efk.hu)
Joaquim J.F. Soares (Joaquim.Soares@ki.se)
Mindaugas Stankunas (stankunas@med.kmu.lt)
Christa Straßmayr (Christa.Strassmayr@lubis.lbg.ac.at)
Marta Welbel (kwzoo@ipin.edu.pl)
Stefan Priebe (s.priebe@qmull.ac.uk)

Version: 7 Date: 6 September 2011

Author's response to reviews:

Cover letter September 6, 2011

Manuscript:
Health care for immigrants in Europe: is there still consensus among country experts about principles of good practice? A Delphi study.

Walter Devillé1*, Tim Greacen2, Marija Bogic3, Marie Dauvrin4, Sónia Dias5, Andrea Gaddini6, Natasja Koitzsch Jensen7, Christine Karamanidou8, Ulrike Kluge9, Ritva Mertaniemi10, Rosa Puigpinós i Riera11, Attila Sárváry12, Joaquim J.F. Soares13, Mindaugas Stankunas14, Christa Straßmayr15, Marta Welbel16, Stefan Priebe3.

Dear editor,

We revised the manuscript making the following formatting changes:

Major revisions:

References: reference 27, Submitted/ unpublished. As this reference will appear as a bookchapter only by the end of this year, we withdrew this reference and
adapted the numbers of the following references.

Tables: shading and colours within the tables have been taken away.

Minor revisions :

Highlighting/tracking: removed

Tables: vertical lines are removed. There are no merged cells in the tables.

Typography: Spelling and errors are checked.

With best regards,

Walter Devillé

Senior researcher

NIVEL